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Device Service Record Form for AES003
	Date
	Comments

(Include the following in each DSR entry: What, Where, Why)
	Info Provided By

	1/23/09
	This is the DSR for the 9-cell cavity AES003.
	Jamie Blowers

	11/4/09
	There are no QC data recorded in Vector for this cavity. This is because we did not have travelers in place when this cavity was first received.
	Jamie Blowers

	Unknown
	EP 177 microns at JLab
	Camille Ginsburg

	6/5/2007
	RF test: 18.7 MV/m (JLab), limitation quench
	Camille Ginsburg

	Unknown
	EP 200 microns (total) at JLab
	Camille Ginsburg

	7/27/2007
	RF test: 17.6 MV/m (JLab), limitation quench
	Camille Ginsburg

	Unknown
	EP at JLab
	Camille Ginsburg

	8/6/2007
	RF test: 17.3 MV/m (JLab), limitation quench
	Camille Ginsburg

	Unknown
	EP at JLab
	Camille Ginsburg

	10/17/2007
	RF test: 21.0 MV/m (JLab), limitation quench
	Camille Ginsburg

	1/28/2008
	RF test: 20.0 MV/m (FNAL), limitation quench
	Camille Ginsburg

	4/3/2008
	RF test 11.0 MV/m (FNAL), quench/FE, variable coupler commissioning

Ozelis test note:AES003 Test Note 
	Camille Ginsburg

	4/29/2008
	Sent to T. Tajima (LANL) for test stand and instrumentation commissioning
	Camille Ginsburg

	10/20/2008
	T. Tajima slides at TTC New Delhi (see 3/1/2009 note below)
	Camille Ginsburg

	12/8/2008
	C.M. Ginsburg AES1-4 performance summary
	Camille Ginsburg

	1/6/2009
	Imprinted serial number confirmed to be AES003 by T. Tajima/G. Erimeev
	Camille Ginsburg

	3/1/2009
	T. Tajima sent this file with the updated results from the cavity inspection and RF test.  These results supersede those shown last October in India (above).
	Camille Ginsburg

	5/7/2009
	This cavity was received from Los Alamos 1/29/09.
	Mark Steinke

	6/4/2009
	Cavity has been moved to IB#4 awaiting shipment to KEK.
	John Zweibohmer

	6/12/2009
	Cavity was received by Hitoshi Hayano at KEK today.  
His plan:

Connect nitrogen in the cavity vacuum valve and pressurize, then open valve slowly.  Then, perform optical inspection and grinding, rinsing.
	Camille Ginsburg

	7/6/2009
	First KEK report on AES003 from Ken Watanabe and Hitoshi Hayano regarding warm RF tests and optical inspections received yesterday.
Hayano-san’s plan:

We observed a lot of scratch on iris, and three major bumps at HAZ of equator EBW seam. However the inspection on the outside of HAZ is not yet done, so that we can not compare the result with T-map and LANL pictures.  Anyway, places for local grinding are expected to so many places. We should prepare the flexible-mover type local grinder (2nd version of local grinder) to be ready. It will take another one or two months.
	Camille Ginsburg

	10/15/09
	Cavity location KEK, confirmed by Camille 10/14/09.
	Jim Rife

	11/3/09
	Cavity delivered to ICB. Status changed from “reserved: offsite” to “reserved: onsite”.
	Jamie Blowers

	11/4/09
	Cavity delivered to MP9 11/4/09.  Status changed from location “ICB” to location “MP9”.  Cavity at MP9 for storage.
	Jim Rife

	11/9/2009
	KEK activity report from Ken Watanabe via Hitoshi Hayano: pdf document ppt document
	Camille Ginsburg

	11/9/2009
	Tsuyoshi Tajima would like to borrow AES003 for T-mapping (without opening the cavity) but we are planning to test the cavity as soon as we can at Fermilab and this proposal is not possible.

	Camille Ginsburg


	12/15/09
	Location verification.  Cavity at MP9 12/15/09. 
	Jim Rife

	1/5/10
	Cavity going to ANL picked up @ 7:30 a.m.
	Mark Chlebek

	1/5/10
	Cavity was delivered to ANL at 9am from MP9
	Jerel Brown

	1/5/10
	Cavity received at ANL at 9AM
	Chris Baker

	1/6/2010
	Damon Bice says that the cavity was not under vacuum.  It seems likely that the vacuum valve was not fully closed.  We will proceed with the single HPR, removing only the bottom flange for the HPR, before the vertical test planned for next week.  The cavity should arrive at IB1 Friday 1/8 or latest Monday 1/11.
	Camille Ginsburg

	1/11/2010
	Cavity’s upper beam line flange was leaking. Flange was tightened, the rechecked and found to be leak tight. Cavity was then transported to IB#1 for VT
	Damon Bice

	1/12/2010
	Cold RF test in IB1:  30.0 MV/m quench limitation accompanied by field emission.  Thermometry indicated no isolated hot spots in the previous localized heating location, so it appears the spot polishing at KEK was proven successful from the test perspective.  HPR at ANL is next, then 120C bake and re-test.

Ozelis test note
	Camille Ginsburg

	1/12/2010
	VTS data filename is unusual: TE9AES003.txt (note of 12/28/12)
	Alex Melnychuk

	1/15/10
	Moved from IB1 to ANL
	L. Zimmerman

	1/22/10
	Cavity was fully disassembled, 2 HPR and reassembled. Transported to IB1 under vacuum
	Damon Bice

	1/22/2010
	120 C bake for 48 hours in IB1.
	Camille Ginsburg

	1/26/2010
	Cold RF test in IB1. 

34.5 MV/m (power limitation),Q0= 4.9E9, at 2K

34.6 MV/m (quench/FE limitation), Q0=5.7E9 at 1.8K

Residual resistance ~6 nOhm

Next: optical inspection, tuning, light EP, re-test

Ozelis test note
	Camille Ginsburg

	2/3/2010
	Cavity was successfully moved from IB1 to ICB @ 1100 hours
	Rick Espinoza

	3/2/10
	A CD with all the AES cavity QC data was received today. I have uploaded the QC report to a “Generic” traveler in Vector.
	Jamie Blowers

	3/16/2010
	Donna Hicks and Dave Burk polished out a stain in cell 1 very close to the beampipe iris today, using an aluminum oxide “sand”-paper.  An initial wiping with alcohol had no effect.  Dmitri did the optical inspection.  Below is the photographic progression of the stain.
@KEK @FNAL: Before Midway After
	Camille Ginsburg

	3/22/2010
	AES003 field flatness measurements by Timer before and after tuning (both are after the 1/26/2010 vertical test): spreadsheet
Before tuning the field flatness (lowest/highest cell) is 87%.  After tuning, it is >98%.  Timer’s estimate is that the reported gradient from the vertical test is a few percent too high.
	Camille Ginsburg

	3/26/10
	Cavity transported to ANL for light EP and Assy sequence
	Damon Bice

	4/2/2010
	Tom Reid spreadsheet shows 20 um light EP on 4/2/2010
	Camille Ginsburg

	4/15/10
	Cavity received average 22.07 microns of EP along with HPR and assy sequence. After leak check, cavity was valved off at 1.5 e-7, crated, and stored  in ANL cavity storage area awaiting VTS readiness
	Damon Bice

	4/29/10
	Cavity transported to VTS
	Damon Bice

	5/4/2010
	Vertical test in IB1: 32.2 MV/m (quench)
Gradient reduced from 34.7 MV/m (8% reduction) in previous test, i.e., is within typical measurement error.  FE onset initially 23 MV/m, but emitter processed away before reaching max gradient.  Mode measurements indicate cells 1, 4, 6 and 9 lowest performers (32-33 MV/m).
Preliminary Ozelis test summary (ppt)
	Camille Ginsburg

	5/17/10
	Cavity transported from IB1 to ICB Awaiting further instruction to proceed
	Damon Bice

	8/5/10
	Cavity transported from ICB to IB1 VTS
	L. Zimmerman / J. Brown

	8/5/10
	Received at IB1
	M carter

	8/6/10
	Plans changed cavity recrated and ready for transport
	M carter

	8/9/2010
	Cavity moved from IB1 to MP9 @ 1330 hours per Camille Ginsburg
	RE / LZ

	8/30/10
	Cavity at IB4 inspections.
	Mark Chlebek

	8/31/10
	Cavity measured on CMM and returned to crate.
	Robert Riley

	8/31/10
	Cavity transported to ANL for HPR / Assy sequence
	Damon Bice

	9/1/10
	During incoming inspection @ ANL, a gouge was noted on the equator of cell #2 (when counting from non FPC end) 
	Damon Bice

	9/9/10
	Cavity received HPR / Assy sequence and was transported to VTS
	Damon Bice

	9/13/10
	VTS test
	Alex Melnychuk

	9/16/10
	VTS test
	Alex Melnychuk

	9/10/10
	Cavity received at IB1
	M carter

	9/20/10
	Cavity vented and crated without cage, ready for pickup to op  inspec 
	M Carter

	9/22/10
	Cavity picked up
	M Carter

	9/22/2010
	Cavity moved from IB1 to ICB Optical Inspection Stand per Camille Ginsburg @ 1045 hours
	RE / JB

	Oct-06-2010
	Optical inspection done. No obvious defect observed at the quench location.
	Dmitri Sergatskov

	9/22/11
	Cavity at ICB.  Cavity is in queue for dressing, no hurry.  Last activity est. material removal = 310 um.  See Ginsburg Cavity Scheduling Mtg. 21-Sept-2011 slides for details.
	Jim Rife per Ginsburg Cavity Scheduling Mtg. 21-Sept-2011

	10/10/11
	Cavity at ICB.  Next: dressing.  See Ginsburg Cavity Scheduling Mtg. 5-Oct-2011 slides for details.
	Jim Rife

	12/13/11
	Cavity moved from ICB to IB4 for RF measurements for Timer
	JB / RE

	12/13/2011
	Cavity AES003 RF QC before dressing. F=1297.31 MHz, FF=95.4%, Eccentricity 0.26 mm. Should be tuned.
	Timergali Khabiboulline

	12/14/2011
	Cavity AES003 tuned to F=1297.33 MHz, FF=99.5%, Ecc=0.26 mm.
	Timergali Khabiboulline

	12/15/2011
	Cavity moved from IB4 to ICB per Timergali @ 1130 hours
	RE / JB

	12/19/11
	Est. material removal = 310 um: 240 um EP @ JLab, 50um @ KEK, 20um @ FNAL/ANL, plus local grinding.  Next: dressing after TB9ACC012 (need FF measurements)
	Jim Rife per Ginsburg Cavity Scheduling Mtg. 14-Dec-2011

	8/21/12
	Cavity location verification.  Cavity @ ICB.
	Jim Rife
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